Author’s response to reviews

Title: No long-term impact of low-energy distal radius fracture on health-related quality of life and global quality of life. a case-control study.

Authors:

Gudrun Rohde (gudrun.rohde@sshf.no)
Glenn Haugeberg (glenn.haugeberg@sshf.no)
Anne Marit Mengshoel (a.m.mengshoel@medisin.uio.no)
Torbjorn Moum (torbjorn.moum@medisin.uio.no)
Astrid K Wahl (a.k.wahl@medisin.uio.no)

Version: 25 Date: 21 August 2009

Author’s response to reviews:

I have made some small typographical changes:

Page 7, line 11 - an extra space has been removed
Page 8, line 2 - an extra e has been added in the word quearionnaire
Page 14, line 10 - an extra space has been removed after GQOL
Page 14, line 12 - an extra space has been removed after retirement
Table 2 A hyphen has been added between health-related
Table 4 A hyphen has been added between health-related

I do not need to see the manuscript once more.